Author’s response to reviews

Title: LONG-TERM VISUAL OUTCOMES AND REHABILITATION IN USHER SYNDROME TYPE II AFTER RETINAL IMPLANT ARGUS II.

Authors:
Maria Iglesias (mariaiglesiasalvarez@gmail.com)
Maria Iglesias (mariaiglesiasalvarez@gmail.com)

Version: 4 Date: 26 Jun 2018

Author's response to reviews:

RESPONSE TO THE COMMENTS

BOPH-D-17-00591R1

LONG-TERM VISUAL OUTCOMES AND REHABILITATION IN USHER SYNDROME TYPE II AFTER RETINAL IMPLANT ARGUS II.

Jeroni Nadal; Maria Iglesias

BMC Ophthalmology

We would like to thank the editor for all the advices and support. We believe that with your comments the quality of the article improves in order to be finally accepted for its publication.

Kind regards,

Maria Iglesias

Jeroni Nadal
Editor Comments:

1. Please provide separate Figure legends for Figure 2 and Figure 3. Figure 4 and Figure 5 could be combined as Figure 4, and please provide the composite figure 4 upon submission.

Response:

Modifications have been made in the text, and a composite figure 4 created and included.

2. "All data generated or analysed during this study are included in this published article [and its supplementary information files]." , please remove the bracket.

Response:

Modified.

3. Please provide us the clean final version without track changes.

Response:

A clean final version of the manuscript is included.